%\documentclass[12pt]{article}
%\usepackage[a4paper,left=30mm,right=20mm,top=20mm,bottom=20mm]{geometry}
%\usepackage{graphics}
%\usepackage{amsmath}
%\usepackage{amssymb}
%\usepackage{epsfig}
%\usepackage{cite}
%\usepackage{xcolor}
%\usepackage[unicode]{hyperref}
%\bibliographystyle{unsrt}
\documentclass[12pt,a4paper]{panl}
\usepackage{cite}
\usepackage{wrapfig}
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{amsmath}
\usepackage{longtable}
\usepackage{rotating}
\usepackage{lscape}
\usepackage{epsfig}
\usepackage{multirow}

\newcommand{\z}{&&\hspace*{-1cm}}
\newcommand{\zz}{&&\hspace*{-4cm}}
\newcommand{\ep}{\varepsilon}
\newcommand{\cita} [1] {$^{\hbox{\scriptsize \cite{#1}}}$}
\newcommand{\prepr}[1] {\begin{flushright} {\bf #1} \end{flushright} \vskip 1.5cm}
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\eea}{\end{eqnarray}}
\newcommand{\be}{\begin{equation}}
\newcommand{\ee}{\end{equation}}
\newcommand{\MSbar}{\overline{\rm MS}}
\newcommand{\as}{\alpha_s}
\newcommand{\asMZ}{\alpha_s(M^2_Z)}
\newcommand{\ar}{a_s}

\originalTeX
%\russianTeX
\begin{document}
\begin{thebibliography}{1}
\def\selectlanguageifdefined#1{
\expandafter\ifx\csname date#1\endcsname\relax
\else\selectlanguage{#1}\fi}
\providecommand*{\href}[2]{{\small #2}}
\providecommand*{\url}[1]{{\small #1}}
\providecommand*{\BibUrl}[1]{\url{#1}}
\providecommand{\BibAnnote}[1]{}
\providecommand*{\BibEmph}[1]{\emph{#1}}
\ProvideTextCommandDefault{\cyrdash}{\hbox to.8em{--\hss--}}
\providecommand*{\BibDash}{\ifdim\lastskip>0pt\unskip\nobreak\hskip.2em\fi
\cyrdash\hskip.2em\ignorespaces}


\bibitem{KL2025}
\selectlanguageifdefined{english}
\BibEmph{Kotikov A.V., Lipatov A.V.} {Updating TMD parton densities in a proton within
  the Kimber-Martin-Ryskin approach}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.111.094009}{Phys.  Rev. D}
  \BibDash
\newblock 2025. \BibDash
\newblock V.~111, no.~9. \BibDash
\newblock P.~094009. \BibDash
\newblock arXiv:2502.19126~[hep-ph].

\bibitem{KL2025res}
\selectlanguageifdefined{english}
\BibEmph{Kotikov A.V., Lipatov A.V.} {A simple model for nuclear modification of parton distribution functions} \BibDash
\newblock 2025. \BibDash
\newblock arXiv:2506.16183~[hep-ph].

\bibitem{MKla24}
\selectlanguageifdefined{english}
\BibEmph{Klasen M., Paukkunen H.} {Nuclear PDFs After the First Decade of LHC Data}~//
  \href{http://dx.doi.org/10.1146/annurev-nucl-102122-022747}{Ann. Rev. Nucl. Part. Sci.}
  \BibDash
\newblock 2024. \BibDash
\newblock V.~74, no.~1. \BibDash
\newblock P.~49--87. \BibDash
\newblock arXiv:2311.00450~[hep-ph].


\bibitem{EMCEffect}
\selectlanguageifdefined{english}
\BibEmph{Aubert J.J. et~al.} [European Muon Collaboration] {The ratio of the nucleon
  structure functions $F2_n$ for iron and deuterium}~//
  \href{http://dx.doi.org/10.1016/0370-2693(83)90437-9}{Phys. Lett. B}
  \BibDash
\newblock 1983. \BibDash
\newblock V.~123. no.~2. \BibDash
\newblock P.~275--278. \BibDash
%\newblock arXiv:2312.16293~[hep-ph].




\bibitem{RescalingModel-1}
\selectlanguageifdefined{english}
\BibEmph{Jaffe R.L., Close F.E., Roberts R.G., Ross G.G.} {On the Nuclear Dependence
  of Electroproduction}~//
  \href{http://dx.doi.org/10.1016/0370-2693(84)91383-2}{Phys. Lett. B}
  \BibDash
\newblock 1984. \BibDash
\newblock V.~134, no.~3. \BibDash
\newblock P.~.449--454 \BibDash
%\newblock arXiv:1706.01849~[hep-ph].

\bibitem{RescalingModel-1b}
\selectlanguageifdefined{english}
\BibEmph{Nachtmann O., Pirner H.J.} {Color Conductivity in Nuclei and the Emc Effect}~//
  \href{http://dx.doi.org/10.1007/BF01577042}{Z. Phys. C}
  \BibDash
\newblock 1984. \BibDash
\newblock V.~21, no.~2. \BibDash
\newblock P.~.277 \BibDash
%\newblock arXiv:1706.01849~[hep-ph]



\bibitem{RescalingModel-1c}
\selectlanguageifdefined{english}
\BibEmph{Close F.E., Jaffe R.L., Roberts R.G., Ross G.G.} {Change of Confinement Scale in Nuclei:
  Predictions for Structure Functions Confront Electroproduction Data}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.31.1004}{Phys.  Rev. D }
  \BibDash
\newblock 1985. \BibDash
\newblock V.~31, no.~5.      . \BibDash
\newblock P.~.`1004    \BibDash
%\newblock arXiv:1706.01849~[hep-ph].

\bibitem{RescalingModel-1d}
\selectlanguageifdefined{english}
\BibEmph{Close F.E., Jaffe R.L., Roberts R.G., Ross G.G.} {Change of Confinement Scale in Nuclei:
  Predictions for Structure Functions Confront Electroproduction Data}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.31.1004}{Phys.  Rev. D }
  \BibDash
\newblock 1985. \BibDash
\newblock V.~31, no.~5.      . \BibDash
\newblock P.~.`1004    \BibDash
%\newblock arXiv:1706.01849~[hep-ph].

\bibitem{RescalingModel-2}
\selectlanguageifdefined{english}
\BibEmph{Close F.E., Roberts R.G., Ross G.G.} {The Effect of Confinement Size on
  Nuclear Structure Function}~//
  \href{http://dx.doi.org/10.1016/0370-2693(83)90679-2}{Phys. Lett. B}
  \BibDash
\newblock 1983. \BibDash
\newblock V.~129, no.~3.      . \BibDash
\newblock P.~. 346--350 \BibDash
%\newblock arXiv:1706.01849~[hep-ph]


\bibitem{RescalingModel-2b}
\selectlanguageifdefined{english}
\BibEmph{Jaffe R.L.} {Quark Distributions in Nuclei}~//
  \href{http://dx.doi.org10.1103/PhysRevLett.50.228/}{Phys. Rev. Lett. }
  \BibDash
\newblock 1983. \BibDash
\newblock V.~50, no.~5.      . \BibDash
\newblock P.~.228    \BibDash
%\newblock arXiv:1706.01849~[hep-ph].




\bibitem{RescalingModel-Apps1}
\selectlanguageifdefined{english}
\BibEmph{Kotikov A.V., Shaikhatdenov B.G., Zhang P.M.} {Application of the rescaling
  model at small Bjorken $x$ values}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.96.114002}{Phys. Rev. D}
  \BibDash
\newblock 2017. \BibDash
\newblock V.~96, no.~11. \BibDash
\newblock P.~114002. \BibDash
\newblock arXiv:1706.01849~[hep-ph].


\bibitem{RescalingModel-Apps2}
\selectlanguageifdefined{english}
\BibEmph{Kotikov A.V., Lipatov A.V., Zhang P.M.} {Shadowing and antishadowing
  in the rescaling model}~//
  \href{http://dx.doi.org/10.31857/S1234567824100021}{JETP Lett.}
  \BibDash
\newblock 2024. \BibDash
\newblock V.~119, no.~10. \BibDash
\newblock P.~729--730. \BibDash
\newblock arXiv:2404.07071~[hep-ph].



\bibitem{AbKoLi23}
\selectlanguageifdefined{english}
\BibEmph{Abdulov N.A., Kotikov A.V., Lipatov A.V.} {Gluon Density in the Rescaling Model}~//
  \href{http://dx.doi.org/10.1134/S1547477123040027}{Phys. Part. Nucl. Lett.}
  \BibDash
\newblock 2023. \BibDash
\newblock V.~20, no.~4. \BibDash
\newblock P.~557--559. \BibDash
\newblock arXiv:2211.14209~[hep-ph].


\bibitem{gDAS1}
\selectlanguageifdefined{english}
\BibEmph{Kotikov A.V., Parente G.} {Small x behavior of parton distributions with
  soft initial conditions}~//
  \href{http://dx.doi.org/10.1016/S0550-3213(99)00107-8}{Nucl. Phys. B}
  \BibDash
\newblock 1999. \BibDash
\newblock V.~549, no.~3. \BibDash
\newblock P.~242--262. \BibDash
\newblock arXiv:hep-ph/9807249~[hep-ph].


\bibitem{Illarionov:2004nw}
\selectlanguageifdefined{english}
\BibEmph{Illarionov A.Y., Kotikov A.V., Parente Bermudez G.} {Small x behavior of parton
  distributions. A Study of higher twist effects}~//
  \href{http://dx.doi.org/10.1134/S1063779608030015}{Phys. Part. Nucl.}
  \BibDash
\newblock 2008. \BibDash
\newblock V.~39, no.~3. \BibDash
\newblock P.~307--347. \BibDash
\newblock arXiv:hep-ph/0402173~[hep-ph].




\bibitem{PDFs-early-approach}
\selectlanguageifdefined{english}
\BibEmph{Lopez C., Yndurain F.J.} {Behavior of Deep Inelastic Structure Functions
  Near Physical Region Endpoints From {QCD}}~//
  \href{http://dx.doi.org/10.1016/0550-3213(80)90369-7}{Nucl. Phys. B }
  \BibDash
\newblock 1980. \BibDash
\newblock V.~171, no.~2. \BibDash
\newblock P.~. 231--252 \BibDash
%\newblock arXiv:1005.3772~[hep-ph].


\bibitem{PDFs-our}
\selectlanguageifdefined{english}
\BibEmph{Abdulov N.A., Kotikov A.V., Lipatov A.V.} {Parametrizations of Collinear and
  k$_{T}$-Dependent Parton Densities in Proton}~//
  \href{http://dx.doi.org/10.3390/particles5040039}{Particles}
  \BibDash
\newblock 2022. \BibDash
\newblock V.~5, no.~4. \BibDash
\newblock P.~535--560. \BibDash
\newblock arXiv:2210.17396~[hep-ph].

\bibitem{PDFs-our-previous}
\selectlanguageifdefined{english}
\BibEmph{Illarionov A.Y., Kotikov A.V., Parzycki S.S., Peshekhonov D.V.} {New type of
  parametrizations for parton distributions}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.83.034014}{Phys. Rev. D}
  \BibDash
\newblock 2011. \BibDash
\newblock V.~83, no.~3. \BibDash
\newblock P.~034014. \BibDash
\newblock arXiv:1005.3772~[hep-ph].

\bibitem{CT14}
\selectlanguageifdefined{english}
\BibEmph{Dulat S., Hou T.J., Gao J., Guzzi M., Huston J., Nadolsky P., Pumplin J.,
  Schmidt C., Stump D., Yuan C.P.} {`New parton distribution functions from a
  global analysis of quantum chromodynamics}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.93.033006}{Phys. Rev. D}
  \BibDash
\newblock 2016. \BibDash
\newblock V.~93, no.~3. \BibDash
\newblock P.~033006. \BibDash
\newblock arXiv:1506.07443~[hep-ph].
%\cite{NNPDF:2021njg}

\bibitem{NNPDF4}
\selectlanguageifdefined{english}
\BibEmph{Ball R.D. et~al.} [NNPDF Collaboration] {The path to proton structure at
  1{\%} accuracy}~//
  \href{http://dx.doi.org/10.1140/epjc/s10052-022-10328-7}{Eur. Phys. J. C}
  \BibDash
\newblock 2022. \BibDash
\newblock V.~82. no.~5. \BibDash
\newblock P.~428. \BibDash
\newblock arXiv:2109.02653~[hep-ph]

\bibitem{MSHT20}
\selectlanguageifdefined{english}
\BibEmph{Bailey S., Cridge T., Harland-Lang L.A.,, Martin A.D., Thorne R.S.}
        {Parton distributions from LHC, HERA, Tevatron and fixed target data: MSHT20 PDFs}~//
  \href{http://dx.doi.org/10.1140/epjc/s10052-021-09057-0}{Eur. Phys. J. C}
  \BibDash
\newblock 2021. \BibDash
\newblock V.~81. no.~4. \BibDash
\newblock P.~341. \BibDash
\newblock arXiv:2012.04684~[hep-ph]


\bibitem{IMP}
\selectlanguageifdefined{english}
\BibEmph{Wang R., Chen X.}
        {Dynamical parton distributions from DGLAP equations with nonlinear corrections}~//
  \href{http://dx.doi.org/10.1088/1674-1137/41/5/053103}{Chin. Phys. C}
  \BibDash
\newblock 2017. \BibDash
\newblock V.~41. no.~5. \BibDash
\newblock P.~053103. \BibDash
\newblock arXiv:1609.01831~[hep-ph]


\bibitem{PDG}
\selectlanguageifdefined{english}
\BibEmph{Navas S. et~al.} [Particle Data Group] {Review of particle physics}~//
  \href{http://dx.doi.org/10.1103/PhysRevD.110.030001}{Phys. Rev. D}
  \BibDash
\newblock 2024. \BibDash
\newblock V.~110. no.~3. \BibDash
\newblock P.~030001. \BibDash
%\newblock arXiv:2312.16293~[hep-ph].


\bibitem{frozen-aQCD}
\selectlanguageifdefined{english}
\BibEmph{Kotikov A.V., Lipatov A.V., Zotov N.P.} {The Longitudinal structure function F(L):
  Perturbative QCD and k(T) factorization versus experimental data at fixed W}~//
  \href{http://dx.doi.org/10.1134/1.2149061}{J. Exp. Theor. Phys.}
  \BibDash
\newblock 2005. \BibDash
\newblock V.~101, no.~5. \BibDash
\newblock P.~811--816. \BibDash
\newblock arXiv:hep-ph/0403135~[hep-ph].

\bibitem{gDAS-PDFs-smallx}
\selectlanguageifdefined{english}
\BibEmph{Cvetic G., Illarionov A.Y., Kniehl B.A., Kotikov A.V.} {Small-x behavior of the
  structure function F(2) and its slope partial ln F(2) / partial ln(1/x) for 'frozen' and analytic strong-coupling constants}~//
  \href{http://dx.doi.org/10.1016/j.physletb.2009.07.057}{Phys. Lett. B}
  \BibDash
\newblock 2009. \BibDash
\newblock V.~679, no.~3. \BibDash
\newblock P.~350--354. \BibDash
\newblock arXiv:0906.1925~[hep-ph].


\bibitem{nIMP}
\selectlanguageifdefined{english}
\BibEmph{Wang R., Chen X.}
        {Global study of nuclear modifications on parton distribution functions}~//
  \href{http://dx.doi.org/10.1016/j.nuclphysb.2017.04.008}{Nucl. Phys. B}
  \BibDash
\newblock 2017. \BibDash
\newblock V.~920. no.~1. \BibDash
\newblock P.~ 1-19. \BibDash
\newblock arXiv:1611.03670~[hep-ph]


\bibitem{nCTEQ}
\selectlanguageifdefined{english}
\BibEmph{Denniston A.W. et~al.} [nCTEQ Collaboration] {Modification of Quark-Gluon Distributions in Nuclei by Correlated Nucleon Pairs}~//
  \href{http://dx.doi.org/10.1103/PhysRevLett.133.152502}{Phys. Rev. Lett}
  \BibDash
\newblock 2024. \BibDash
\newblock V.~133. no.~15. \BibDash
\newblock P.~152502. \BibDash
\newblock arXiv:2312.16293~[hep-ph].

\bibitem{nNNPDF3}
\selectlanguageifdefined{english}
\BibEmph{Abdul Khalek R., Gauld R., Giani T., Nocera E.R., Rabemananjara T.R., Rojo J.}
        {nNNPDF3.0: evidence for a modified partonic structure in heavy nuclei}~//
  \href{http://dx.doi.org/10.1140/epjc/s10052-022-10417-7}{Eur. Phys. J. C }
  \BibDash
\newblock 2022. \BibDash
\newblock V.~82. no.~6. \BibDash
\newblock P.~507. \BibDash
\newblock arXiv:2201.12363~[hep-ph].


\bibitem{EPPS21}
\selectlanguageifdefined{english}
\BibEmph{Eskola K.J., Paakkinen P., Paukkunen H., Salgado C.A.}
        {EPPS21: a global QCD analysis of nuclear PDFs}~//
  \href{http://dx.doi.org/10.1140/epjc/s10052-022-10359-0}{Eur. Phys. J. C }
  \BibDash
\newblock 2022. \BibDash
\newblock V.~82. no.~5. \BibDash
\newblock P.~413. \BibDash
\newblock arXiv:2112.12462~[hep-ph].

\end{thebibliography}

% Journal sections (see
%http://pkp.jinr.ru/index.php/PEPAN_LETTERS/about/editorialPolicies#focusAndScope)
%\issuearea{Physics of Elementary Particles and Atomic Nuclei. Theory}
% or in Russian
%\issuearea{ФИЗИКА ЭЛЕМЕНТАРНЫХ ЧАСТИЦ И АТОМНОГО ЯДРА. ТЕОРИЯ}

\title{A simple model for PDFs and nPDFs}
\maketitle
\authors{A.V.~Kotikov$^{a,}$\footnote{E-mail:  kotikov@theor.jinr.ru},
A.V.~Lipatov$^{b,}$\footnote{E-mail: artem.lipatov@mail.ru}}
\setcounter{footnote}{0}
\from{$^{a}$\,BLTP, Joint Institute for Nuclear Research, Dubna 141980, Moscow
region, Russia}
\from{$^{b}$\,Skobeltsyn Institute of Nuclear Physics, Lomonosov Moscow State
University, 119991 Moscow, Russia}

\begin{abstract}
We present the main results of our recent papers \cite{KL2025,KL2025res}. In
\cite{KL2025}, we derived an analytical solution of the QCD evolution equations for
parton distribution functions.
%(PDFs).
The valence and non-singlet quark components
satisfy the Gross-Llewellyn-Smith and Gottfried sum rules, respectively, while
momentum conservation is maintained for the singlet quark and gluon densities.
Several phenomenological parameters were extracted from a combined fit to precision
data on the proton structure function $F_2(x,Q^2)$ collected by the BCDMS, H1, and
ZEUS Collaborations, comprising a total of 933 points from 5 datasets. In
\cite{KL2025res}, we proposed a model for nuclear medium modifications of parton
densities. The approach is based on a global analysis of available deep inelastic
scattering data for different nuclear targets within the rescaling model,
incorporating Fermi motion effects. By fitting the rescaling parameters to
experimental data on the ratio $F_2^A(x,Q^2)/F_2^{D}(x,Q^2)$ for several nuclear
targets $A$, we obtained predictions for nuclear parton distribution functions.
%(nPDFs).
\end{abstract}
\vspace*{6pt}

\noindent
PACS: 13.60.Le, 13.85.-t, 11.10.Ef,  12.40.Nn

\section*{Introduction} \indent

The study of deep inelastic scattering (DIS) of leptons on nuclei reveals
significant effects of nucleon interactions within the nucleus, challenging the
naive picture of the nucleus as a system of quasi-free nucleons (see the recent
review \cite{MKla24}). Nuclear medium effects on parton distribution functions
(PDFs) attract considerable interest from both experimental and theoretical
perspectives. A detailed understanding of nuclear modifications of parton densities
(nPDFs) is a fundamental problem in nuclear and high-energy physics, crucial for
theoretical descriptions of $pA$ and $AA$ collisions at modern (LHC, RHIC) and
future colliders (FCC-he, EiC, EicC, CEPC, NICA). The nuclear modification factor is
typically defined as the ratio of per-nucleon DIS structure functions in nuclei $A$
and deuteron $D$, $R^A = F_2^A(x,Q^2)/F_2^D(x,Q^2)$, or equivalently, the ratio of
corresponding parton densities. This ratio exhibits distinct behaviors across
different kinematic regions: shadowing ($x \leq 0.1$), anti-shadowing ($0.1 \leq x
\leq 0.3$), valence quark dominance ($0.3 \leq x \leq 0.7$), and Fermi motion ($x
\geq 0.7$). Shadowing and anti-shadowing correspond to $R^A < 1$ and $R^A > 1$,
respectively, while the EMC effect \cite{EMCEffect} and Fermi motion refer to the
slope of $R^A$ in the valence-dominant region and the rise of $R^A$ at large $x$.

Note that the nuclear medium effect was first discovered \cite{EMCEffect} by the
European Muon Collaboration (EMC) in the valence quark dominance region.
%Investigations of shadowing and anti-shadowing (see \cite{Stodolsky, NiZa75,
%ColorDipoleModel-3}) began prior to the availability of EMC experimental data
%\cite{EMC-CCa1, EMC-CCa2} (see also \cite{Ni81} for an overview).

Within the framework of the $Q^2$-rescaling model (hereafter referred to as the
rescaling model), nuclear medium modifications have been investigated
\cite{RescalingModel-1,RescalingModel-1b,RescalingModel-1c,RescalingModel-1d,RescalingModel-2,RescalingModel-2b,RescalingModel-Apps1,RescalingModel-Apps2}.
%\cite{RescalingModel-1,RescalingModel-Apps2}.
%\cite{RescalingModel-1, RescalingModel-Apps1, RescalingModel-Apps2}.
Initially
proposed for the valence-dominant region, this model suggests that the effective
confinement size of gluons and quarks in the nucleus is larger than in the free
nucleon. This shift in the confinement scale leads to a relation between PDFs and
nPDFs through a simple rescaling of their arguments \cite{RescalingModel-1,
RescalingModel-2}. Recently, the rescaling model has been extended to the low-$x$
region \cite{RescalingModel-Apps1}. It has been shown \cite{RescalingModel-Apps1,
AbKoLi23, RescalingModel-Apps2} that good agreement with available experimental data
on $F_2^A(x,Q^2)/F_2^D(x,Q^2)$ ratios at low and moderate $x$ can be achieved by
fitting the corresponding rescaling parameters.

The aim of this work is to present the results of our recent papers
\cite{KL2025,KL2025res}. First, in \cite{KL2025}, we derived analytical expressions
for PDFs from the solution of the DGLAP evolution equations at leading order (LO) in
the QCD coupling. These expressions are based on exact asymptotics at small and
large $x$ (see, e.g., \cite{gDAS1,Illarionov:2004nw} and \cite{PDFs-early-approach}, respectively) and
include subasymptotic terms constrained by momentum conservation and/or the
Gross-Llewellyn-Smith and Gottfried sum rules. Several phenomenological parameters
were determined from a rigorous fit to precision BCDMS, H1, and ZEUS experimental
data on the proton structure function $F_2(x,Q^2)$ over a wide kinematic region: $2
\cdot 10^{-5} \leq x \leq 0.75$ and $1.2 \leq Q^2 \leq 30000$ GeV$^2$. In
\cite{KL2025res}, we employed the solution for PDFs from \cite{KL2025} and used the
rescaling model \cite{RescalingModel-1,RescalingModel-2} to study their nuclear modifications,
additionally incorporating Fermi motion effects. To extract the rescaling
parameters, we performed a global fit to available experimental data on the ratios
$F_2^A(x,Q^2)/F_2^{D}(x,Q^2)$ for different nuclear targets $A$ collected by the
EMC, NMC, SLAC, BCDMS, JLab, and CLAS Collaborations (see \cite{KL2025res} and
references therein).

\section*{Structure function $F_2(x,Q^2)$ and PDFs in a proton} \indent

We begin with some basic formulas used in our calculations. It is well known that
the proton structure function $F_2(x,Q^2)$ at LO in the QCD coupling can be
expressed as:
\begin{gather}
  F_2(x,Q^2) = \sum_{i=1}^{N_f} e_i^2 \left[f_{q_i}(x,Q^2) + f_{\bar{q}_i}(x,Q^2)
\right],
\label{S1.14}
\end{gather}
\noindent
where $e_i$ is the fractional electric charge of quark $q_i$, $N_f$ is the number of
active quark flavors, and $f_{q_i}(x,Q^2)$ and $f_{\bar q_i}(x,Q^2)$ represent the
quark and antiquark densities in a proton (multiplied by $x$), respectively. In the
fixed-flavor-number scheme (FFNS) with $N_f = 4$, where $b$ and $t$ quarks are
%treated
separated out, we have:
\begin{gather}
  F_2(x,Q^2) = \frac{5}{18} \, f_{SI}(x,Q^2) + \frac{1}{6} f_{NS}(x,Q^2),
\label{S1.15}
\end{gather}
\noindent
where the singlet part $f_{SI}(x,Q^2)$ includes the valence and sea quark components:
\begin{gather}
        f_{V}(x,Q^2) = f_u^V(x,Q^2) + f_d^V(x,Q^2), \quad f_{S}(x,Q^2) = \sum_{i=1}^{4}
\left[f_{q_i}^S(x,Q^2) + f_{\bar{q}_i}^S(x,Q^2) \right], \nonumber \\
        f_{SI}(x,Q^2) = \sum_{i=1}^{4} \left[f_{q_i}(x,Q^2) + f_{\bar{q}_i}(x,Q^2) \right]
= f_{V}(x,Q^2) + f_{S}(x,Q^2).
        \label{eq-FSI}
\end{gather}
\noindent
The nonsinglet part $f_{NS}(x,Q^2)$ contains the difference between up and down quarks:
\begin{gather}
    \hspace{-0.5cm}    f_{NS}(x,Q^2) = \sum_{q=u,\,c} \left[f_{q}(x,Q^2) + f_{\bar{q}}(x,Q^2) \right] -
\sum_{q=d,\,s} \left[f_{q}(x,Q^2) + f_{\bar{q}}(x,Q^2) \right].
        \label{eq-FNS-NF4}
\end{gather}
\noindent

These formulas are useful for the accurate determination of phenomenological
parameters in the PDF parameterizations derived in our papers
\cite{KL2025,PDFs-our}. The approach \cite{KL2025,PDFs-our} follows the idea
presented in \cite{PDFs-early-approach,PDFs-our-previous} and consists of two main
steps. First, we analytically compute the asymptotics of the solutions to the DGLAP
equations for parton densities at low and high values of the Bjorken variable $x$.
Second, to obtain analytical expressions for PDFs over the entire $x$ range, we
combine these solutions and interpolate between them.

\section*{Model of nuclear modifications} \indent

As a model for nuclear modifications of parton densities, we consider a combination
of the rescaling model \cite{RescalingModel-1,RescalingModel-2} and Fermi motion. The rescaling model
establishes a simple relationship between ordinary PDFs and nPDFs through a shift in
the kinematic variable $Q^2$.
%(see also \cite{RescalingModel-2}).
For a nucleus $A$,
the valence and nonsinglet parts are modified as:
\begin{gather}
  f_{i}^A(x,Q^2) = f_{i}(x,Q^2_{A,\,i}),
  \label{va.1a}
\end{gather}
\noindent
where $i = V$ or $NS$, and the scale $Q^2_{A,\,i}$ is related to $Q^2$ by:
\begin{gather}
        s^A_i \equiv \ln \left(\frac{\ln\left(Q^2_{A,\,i}/\Lambda^2_{\rm
QCD}\right)}{\ln\left(Q^2_{0}/\Lambda^2_{\rm QCD}\right)}\right) = s +\ln
\left(1+\delta^A_i \right),
        \label{sA}
\end{gather}
\noindent
where $\delta^A_i$ are scale-independent free parameters.

\section*{Numerical results} \indent

In Ref. \cite{KL2025}, we began by determining the phenomenological parameters in
the derived analytical expressions for PDFs in a proton. We then studied their
nuclear modifications in \cite{KL2025res}.

{\bf 1.}~The analytical expressions for PDFs obtained in \cite{KL2025} contain
several parameters that are not predicted by theory. Some of these parameters,
essential in the large-$x$ region, were determined by direct comparison with
numerical solutions of the DGLAP equations at the starting scale $Q_0^2$. We used
the CT'14 (LO) parameterizations \cite{CT14} from the CTEQ-TEA group with $N_f = 4$,
which matches our setup. The remaining parameters were determined by fitting precise
data on the proton structure function $F_2(x,Q^2)$ from the BCDMS, H1, and ZEUS
Collaborations (see \cite{KL2025} and references therein). These data cover an
extremely wide range of $x$ and $Q^2$: $2 \cdot 10^{-5} < x < 0.75$ and $1.2 < Q^2 <
30000$ GeV$^2$, allowing us to extract all necessary parameters simultaneously. Our
fit, comprising a total of 933 points from 5 datasets, achieves an excellent
goodness-of-fit, with $\chi^2/d.o.f. = 1.408$.

\begin{figure}
\begin{center}
\includegraphics[width=6.7cm]{PDFs-Q2=1,0.pdf}
%\includegraphics[width=7.9cm]{Figures/PDFs-Q2=10,0.pdf}
\includegraphics[width=6.7cm]{PDFs-Q2=100,0.pdf}
%\includegraphics[width=7.9cm]{Figures/PDFs-Q2=1000,0.pdf}
\caption{Proton PDFs as functions of $x$ for different values of $Q^2$. For
comparison, we show the results of numerical solutions of the DGLAP equations from
the CTEQ-TEA \cite{CT14}, NNPDF4.0 \cite{NNPDF4}, MSHT'2020 \cite{MSHT20}, and IMP
\cite{IMP} groups.}
\label{fig2}
\end{center}
\end{figure}

Some of the derived PDFs are shown in Fig.~\ref{fig2} as functions of $x$ for
several scales $Q^2$, namely $Q^2 = 1$ and $100$ GeV$^2$. Additionally, we compare
our predictions with the corresponding FFNS results from the CTEQ-TEA Collaboration
\cite{CT14} at $N_f = 4$ and the variable-flavor-number scheme (VFNS) predictions
from the NNPDF4.0 \cite{NNPDF4}, MSHT'2020 \cite{MSHT20}, and IMP \cite{IMP} groups.
One can observe significant discrepancies between the different PDFs, especially at
low $Q^2 \sim 1$ GeV$^2$. Nevertheless, at moderate and higher $Q^2$, we find
reasonable agreement between our analytical results and the relevant numerical
analyses.

{\bf 2.} To extract the rescaling parameters
% $\delta^A_V$, $\delta^A_{NS}$, and $\delta^A_\pm$
for various nuclear targets, in \cite{KL2025res} we performed a
global fit to structure function ratio data from the EMC, NMC, SLAC, BCDMS, E665,
JLab, and CLAS Collaborations (see \cite{KL2025res} and references therein). We
imposed kinematic cuts on the experimental data to ensure they lie in the deep
inelastic region: $Q^2 \geq 1$ GeV$^2$ and $W^2 \geq 4$ GeV$^2$. Following
\cite{KL2025}, we set $\Lambda_{\rm QCD}^{(4)} = 118$ MeV, corresponding to the
world average $\alpha_s(M^2_Z) = 0.1180$ \cite{PDG}. We applied a 'frozen' treatment
of the QCD coupling in the infrared region (see, for example, \cite{frozen-aQCD,
gDAS-PDFs-smallx} and references therein), where $\alpha_s(Q^2) \to \alpha_s(Q^2 +
M_\rho^2)$ with $M_\rho \sim 1$ GeV. This treatment leads to a good description of
the data on the proton structure function $F_2(x,Q^2)$ \cite{KL2025}.

\begin{figure}
        \begin{center}
                \includegraphics[width=5.7cm]{R-pdfs-He4.pdf}
                %\includegraphics[width=4.7cm]{R-pdfs-C12.pdf}
                %\includegraphics[width=4.7cm]{R-pdfs-Ca40.pdf}
                %\includegraphics[width=4.7cm]{R-pdfs-Fe56.pdf}
                %\includegraphics[width=4.7cm]{R-pdfs-Sn118.pdf}
                \includegraphics[width=5.7cm]{R-pdfs-Pb208.pdf}
                \caption{Predicted nuclear modification factors for parton distributions in
several nuclear targets. The results for gluon nuclear modification predicted by
the nIMP group \cite{nIMP} are shown for comparison.}
                \label{fig:3}
        \end{center}
\end{figure}

Using the analytical expressions for the nucleon target from \cite{KL2025},
Eqs.~(\ref{va.1a}) and~(\ref{sA}) for nuclear targets, and the fitted values of the
rescaling parameters (we considered three different forms of $A$-dependence), we can
predict nuclear modifications of PDFs. Our results are shown in Fig.~\ref{fig:3} for
several nuclei. As is well known, nuclear modifications are weakly dependent on
$Q^2$ at low and moderate $Q^2$, so we show only the results for $Q^2 = 10$ GeV$^2$.
For comparison, we also plot the results for gluon nuclear modification factors
predicted by the nIMP group \cite{nIMP}. We find that the shadowing effect for
gluons is generally weaker than for quarks, consistent with other studies (see, e.g.
\cite{nIMP, nCTEQ, nNNPDF3, EPPS21}). Nevertheless, the differences between
these results are rather small, and the predicted shadowing effects are close to the
nIMP expectations. This contrasts with the large-$x$ region, where we find a
significant discrepancy with the nIMP calculations. The latter show very weak
anti-shadowing at $x \sim 0.1$, while our calculations indicate significant
anti-shadowing effects for quarks. Note that other groups \cite{nNNPDF3, EPPS21}
predict very strong anti-shadowing for nuclear gluon densities. Thus, it is
currently difficult to draw specific conclusions. Again, precision measurements at
future colliders (EiC, EicC) are needed to clarify this point.

\section*{Conclusion} \indent

We have presented a brief overview of the results obtained in our recent papers
\cite{KL2025,KL2025res}. In \cite{KL2025}, we derived an analytical solution of the
DGLAP equations for PDFs in a proton, where the valence and non-singlet quark
components satisfy the Gross-Llewellyn-Smith and Gottfried sum rules, and momentum
conservation for the singlet quark and gluon densities is maintained. In
\cite{KL2025res}, we studied nuclear medium modifications of parton densities based
on the rescaling model. Furthermore, we extended the rescaling model to include
Fermi motion effects, providing a consistent description of nuclear modifications
for DIS structure functions and PDFs across the entire kinematic range. Using simple
analytical formulas for proton PDFs derived at LO in the QCD coupling, we performed
a global analysis of available deep inelastic data for different nuclear targets and
extracted the corresponding rescaling parameters. Our results highlight distinct
shadowing and anti-shadowing behaviors for gluons and quarks.

FUNDING\\
This research was supported by the Russian Science
Foundation grant No. 25-22-00066,   https://rscf.ru/en/project/25-22-00066/.
%This work was supported by ongoing institutional funding. No additional grants to carry out
%or direct this particular research were obtained.

CONFLICT OF INTEREST\\
The authors of this work declare that they have no conflicts of interest.

% \section*{Acknowledgements} \indent
%\vskip -0.3cm
ACKNOWLEDGMENTS\\
We thank S.P.~Baranov, H.~Jung, M.A. Malyshev, and N.N. Nikolaev for their interest,
valuable comments, and remarks.
%This research was supported by the Russian Science
%Foundation grant No. 25-22-00066,\\ https://rscf.ru/en/project/25-22-00066/.
%\bibliographystyle{pepan}
%%\bibliography{pepan_biblio}
%\bibliography{EMCiN}
%%\bibliography{KMR}



\end{document}
